Author’s response to reviews

Title: Test of a Workforce Development Intervention to Expand Opioid Use Disorder Treatment Pharmacotherapy Prescribers: Protocol for a Cluster-Randomized Trial

Authors:

Todd Molfenter (todd.molfenter@chess.wisc.edu)
Hannah Knudsen (hkknud2@uky.edu)
Randy Brown (randy.brown@fammed.wisc.edu)
Nora Jacobson (najacobson@wisc.edu)
Julie Horst (jhorst@wisc.edu)
Mark Van Etten (mark.vanetten@wisc.edu)
Jee-Seon Kim (jeeseonkim@wisc.edu)
Eric Haram (eharam@gmail.com)
Elizabeth Collier (Elizabeth.Collier@dhs.wisconsin.gov)
Sanford Starr (sanford.starr@mha.ohio.gov)
Alexander Toy (ajtoy@wisc.edu)
Lynn Madden (lmadden@aptfoundation.org)

Version: 1 Date: 26 Oct 2017

Author’s response to reviews:

Dear Dr. Wilson,

Thank you for the opportunity to revise and resubmit the manuscript entitled: "Test of a Workforce Development Intervention to Expand Opioid Use Disorder Treatment Pharmacotherapy Prescribers: Protocol for a Cluster-Randomized Trial" (IMPS-D-17-00526). The author team appreciated your suggestions and has incorporated your feedback. Below we have provided responses to the reviewers’ comments.

Response to Reviewers
Request) We would appreciate a more detailed description of the PRB components and how they relate to current evidence.

Response) Pages 9 through 11: We have provided a more detailed description of PRB elements and included references to current evidence. Additional references are detailed on pages 22 through 24. Thank you for this suggestion.

There was only one request.